Author's response to reviews

Title: Association between skin diseases and severe bacterial infections in children: case-control study

Authors:

Robbert SA Mohammedamin (s.mohammedamin@erasusmc.nl)
Johannes C van der Wouden (j.vanderwouden@erasusmc.nl)
Sander Koning (s.koning@erasusmc.nl)
Sten P Willemsen (s.willemsen@erasusmc.nl)
Roos MD Bernsen (rdbernsen@gmail.com)
Francois G Schellevis (f.schellevis@nivel.nl)
Lisette WA van Suijlekom-Smit (l.vansuijlekom@erasusmc.nl)
Bart W Koes (b.koes@erasusmc.nl)

Version: 4 Date: 8 August 2006

Author's response to reviews:

In addition, we have removed all the lines that were used to show changes with a previous version.